Author's response to reviews

Title: Implementation of workflow engine technology to deliver basic clinical decision support functionality

Authors:

Vojtech Huser (vojtech.huser@gmail.com)
Luke V Rasmussen (rasmussen.luke@marshfieldclinic.org)
Ryan Oberg (oberg.ryan@marshfieldclinic.org)
Justin B Starren (starren.justin@marshfieldclinic.org)

Version: 4 Date: 24 March 2011

Author's response to reviews:

(plain text cover letter)
(last set of editorial comments included)
(Changes made to manuscript are described in lines with prefix A: (means A:answer))

-------------------------------------------

Major revisions (we require the author to make these changes)

Abstract: please ensure that the Abstract present in the manuscript matches exactly what is in the submission system
A: Updated in submission system (AAA). Now they should match.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Old title: please remove this
A: Removed

Affiliations: Please remember to include the country in the affiliation details
A: Countries added, affiliation updated in one author (J. Starren)

Please remove the lines from around the abstract
A: Removed

Tracking: please remove all tracking from the manuscript
A: all tracking removed (AAA)

Tables: the table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should
Table titles were already short. The revised manuscript has a corrected order of elements: (1) table title, (2) table itself, and (3) table legend.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

A: The final proofing was done.